Author's response to reviews

Title: Costs and effects of a 'healthy living' approach to community development in two deprived communities: findings from a mixed methods study

Authors:

Helen A Snooks (h.a.snooks@swansea.ac.uk)
Bridie A Evans (b.a.evans@swansea.ac.uk)
David Cohen (dcohen@ glam.ac.uk)
Michael Nugent (postmaster@3sf.co.uk)
Frances L Rapport (f.l.rapport@swansea.ac.uk)
Jon Skone (jon.skone@pembrokeshire.gov.uk)
Angie Meredith (angie.mered@fsmail.net)
Tricia Davies (wellsbourne@aol.com)
Diana O'Sullivan (diana.osullivan@pembrokeshire.gov.uk)

Version: 4 Date: 16 December 2010

Author's response to reviews: see over
Dear Sirs/Madams

1866082081366577 - Costs and effects of a 'healthy living' approach to community development aimed at reducing health inequalities in two deprived communities: findings from a mixed methods study

Thank you for your recent correspondence confirming your acceptance of our manuscript for publication in BMC Public Health and asking for formatting revisions.

I confirm the following changes have been made:

Author list: the list of authors in the manuscript is written exactly as they are in the submission system

Abstract: the abstract present in the manuscript matches exactly what is in the submission system

Boxes: these have been removed and the information has been incorporated within the text of the paper

Tables: vertical text has been removed and made horizontal

Affiliations: full capitals are not used for country names which are simply written with a capital letter at the start. There are no abbreviations of country names

Highlighting: all highlighting has been removed from the manuscript and the text is all in black

Authors’ contributions: this section is formatted into a paragraph

Figures: there is minimal white space around the figure

Tables: the visible vertical lines have been removed from the tables

Typography: we have proof read the document for typographical errors

Thank you for the opportunity to format our paper.

Yours sincerely,

B A Evans